Author’s response to reviews

Title: Evaluation of the Performance of Routine Information System Management (PRISM) framework: evidence from Uganda

Authors:

David R Hotchkiss (hotchkis@tulane.edu)
Anwer Aqil (anwer_aqil@isi.com)
Theo Lippeveld (theo_lippeveld@isi.com)
Edward Makooyo (emukooyo@yahoo.com)

Version: 4 Date: 29 June 2010

Author’s response to reviews:

Dear BMC Series Editorial Production Team,

We have made all of the requested changes to the format of the manuscript, as described in your 28 June 2010 email. Please note that the words "Performance of Routine Information System (PRISM)" should be capitalized in the title of the paper. Please let us know if you have any additional formatting requests, or questions.

Best regards,

David Hotchkiss